\documentclass[twocolumn,twoside]{article}
\makeatletter\if@twocolumn\PassOptionsToPackage{switch}{lineno}\else\fi\makeatother

\usepackage{amsfonts,amssymb,amsbsy,latexsym,amsmath,tabulary,graphicx,times,xcolor}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Following additional macros are required to function some 
% functions which are not available in the class used.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\usepackage{url,multirow,morefloats,floatflt,cancel,tfrupee}
\makeatletter


\AtBeginDocument{\@ifpackageloaded{textcomp}{}{\usepackage{textcomp}}}
\makeatother
\usepackage{colortbl}
\usepackage{xcolor}
\usepackage{pifont}
\usepackage[nointegrals]{wasysym}
\urlstyle{rm}
\makeatletter

%%%For Table column width calculation.
\def\mcWidth#1{\csname TY@F#1\endcsname+\tabcolsep}

%%Hacking center and right align for table
\def\cAlignHack{\rightskip\@flushglue\leftskip\@flushglue\parindent\z@\parfillskip\z@skip}
\def\rAlignHack{\rightskip\z@skip\leftskip\@flushglue \parindent\z@\parfillskip\z@skip}

%Etal definition in references
\@ifundefined{etal}{\def\etal{\textit{et~al}}}{}


%\if@twocolumn\usepackage{dblfloatfix}\fi
\usepackage{ifxetex}
\ifxetex\else\if@twocolumn\@ifpackageloaded{stfloats}{}{\usepackage{dblfloatfix}}\fi\fi

\AtBeginDocument{
\expandafter\ifx\csname eqalign\endcsname\relax
\def\eqalign#1{\null\vcenter{\def\\{\cr}\openup\jot\m@th
  \ialign{\strut$\displaystyle{##}$\hfil&$\displaystyle{{}##}$\hfil
      \crcr#1\crcr}}\,}
\fi
}

%For fixing hardfail when unicode letters appear inside table with endfloat
\AtBeginDocument{%
  \@ifpackageloaded{endfloat}%
   {\renewcommand\efloat@iwrite[1]{\immediate\expandafter\protected@write\csname efloat@post#1\endcsname{}}}{\newif\ifefloat@tables}%
}%

\def\BreakURLText#1{\@tfor\brk@tempa:=#1\do{\brk@tempa\hskip0pt}}
\let\lt=<
\let\gt=>
\def\processVert{\ifmmode|\else\textbar\fi}
\let\processvert\processVert

\@ifundefined{subparagraph}{
\def\subparagraph{\@startsection{paragraph}{5}{2\parindent}{0ex plus 0.1ex minus 0.1ex}%
{0ex}{\normalfont\small\itshape}}%
}{}

% These are now gobbled, so won't appear in the PDF.
\newcommand\role[1]{\unskip}
\newcommand\aucollab[1]{\unskip}
  
\@ifundefined{tsGraphicsScaleX}{\gdef\tsGraphicsScaleX{1}}{}
\@ifundefined{tsGraphicsScaleY}{\gdef\tsGraphicsScaleY{.9}}{}
% To automatically resize figures to fit inside the text area
\def\checkGraphicsWidth{\ifdim\Gin@nat@width>\linewidth
	\tsGraphicsScaleX\linewidth\else\Gin@nat@width\fi}

\def\checkGraphicsHeight{\ifdim\Gin@nat@height>.9\textheight
	\tsGraphicsScaleY\textheight\else\Gin@nat@height\fi}

\def\fixFloatSize#1{}%\@ifundefined{processdelayedfloats}{\setbox0=\hbox{\includegraphics{#1}}\ifnum\wd0<\columnwidth\relax\renewenvironment{figure*}{\begin{figure}}{\end{figure}}\fi}{}}
\let\ts@includegraphics\includegraphics

\def\inlinegraphic[#1]#2{{\edef\@tempa{#1}\edef\baseline@shift{\ifx\@tempa\@empty0\else#1\fi}\edef\tempZ{\the\numexpr(\numexpr(\baseline@shift*\f@size/100))}\protect\raisebox{\tempZ pt}{\ts@includegraphics{#2}}}}

%\renewcommand{\includegraphics}[1]{\ts@includegraphics[width=\checkGraphicsWidth]{#1}}
\AtBeginDocument{\def\includegraphics{\@ifnextchar[{\ts@includegraphics}{\ts@includegraphics[width=\checkGraphicsWidth,height=\checkGraphicsHeight,keepaspectratio]}}}

\DeclareMathAlphabet{\mathpzc}{OT1}{pzc}{m}{it}

\def\URL#1#2{\@ifundefined{href}{#2}{\href{#1}{#2}}}

%%For url break
\def\UrlOrds{\do\*\do\-\do\~\do\'\do\"\do\-}%
\g@addto@macro{\UrlBreaks}{\UrlOrds}



\edef\fntEncoding{\f@encoding}
\def\EUoneEnc{EU1}
\makeatother
\def\floatpagefraction{0.8} 
\def\dblfloatpagefraction{0.8}
\def\style#1#2{#2}
\def\xxxguillemotleft{\fontencoding{T1}\selectfont\guillemotleft}
\def\xxxguillemotright{\fontencoding{T1}\selectfont\guillemotright}

\newif\ifmultipleabstract\multipleabstractfalse%
\newenvironment{typesetAbstractGroup}{}{}%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\AtBeginDocument{\let\citet\cite\let\citep\cite}
\usepackage[numbers,sort&compress,numbers,sort&compress]{natbib}



\newif\ifnokeywords\nokeywordsfalse
\def\abbrvJournalTitle{Future J. Pharm. Health. Sci.}
\def\jrnlurlname{fjphs}
\usepackage{balance}

\makeatletter\input{size10-pointfive.clo}\makeatother%
\definecolor{kwdboxcolor}{RGB}{242,242,242}
\usepackage[hidelinks,colorlinks=true,allcolors=blue]{hyperref}
\linespread{1}                 
\def\floatpagefraction{0.8}

\usepackage[paperheight=11.69in,paperwidth=8.26in,top=1in,bottom=1in,left=1in,right=.75in,headsep=24pt]{geometry}

\usepackage{multirow-custom}

\makeatletter

\def\hlinewd#1{%
  \noalign{\ifnum0=`}\fi\hrule \@height #1%
  \futurelet\reserved@a\@xhline}
\def\tbltoprule{\hlinewd{1pt}\\[-14pt]}
\def\tblbottomrule{\noalign{\vspace*{6pt}}\hline\noalign{\vspace*{2pt}}}
\def\tblmidrule{\hline\noalign{\vspace*{2pt}}}



\let\@articleType\@empty
\let\@journalDoi\@empty
\let\@journalVolume\@empty
\let\@journalIssue\@empty
\let\@crossMarkLink\@empty
\let\@receivedDate\@empty
\let\@acceptedDate\@empty
\let\@revisedDate\@empty
\let\@publishedDate\@empty
\let\@copyrightYear\@empty
\let\@firstPage\@empty
\let\@eISSN\@empty
\let\@pISSN\@empty
\def\eISSN#1{\gdef\@eISSN{#1}}
\def\pISSN#1{\gdef\@pISSN{#1}}
\def\articleType#1{\gdef\@articleType{#1}}
\def\journalDoi#1{\gdef\@journalDoi{#1}}
\def\crossMarkLink#1{\gdef\@crossMarkLink{#1}}
\def\receivedDate#1{\gdef\@receivedDate{#1}}
\def\acceptedDate#1{\gdef\@acceptedDate{#1}}
\def\revisedDate#1{\gdef\@revisedDate{#1}}
\def\publishedDate#1{\gdef\@publishedDate{#1}}
\def\copyrightYear#1{\gdef\@copyrightYear{#1}}
\def\journalVolume#1{\gdef\@journalVolume{#1}}
\def\journalIssue#1{\gdef\@journalIssue{#1}}
\def\firstPage#1{\gdef\@firstPage{#1}}


\def\author#1{%
  \gdef\@author{%
    \hskip-\dimexpr(\tabcolsep)\hskip5pt%
    \parbox{\dimexpr\textwidth-1pt}%
    {\fontsize{11}{13}\selectfont\raggedright #1}%
  }%
}


\usepackage{pharmascope-abs}
\usepackage{caption}
\usepackage{lastpage}
\usepackage{fancyhdr}
\usepackage[noindentafter,explicit]{titlesec}
\usepackage{fontspec}

\setmainfont[%
BoldFont=cambriab.ttf,%
ItalicFont=CAMBRIAI.TTF,%
BoldItalicFont=CAMBRIAZ.TTF]{Cambria.ttf}

\usepackage{qrcode}

%\def\abbrvJournalTitle{Future J. Pharm. Health. Sci.}
\def\jshort{fjphs}

\def\title#1{%
  \gdef\@title{%
%      \vspace*{-20pt}%
       \vspace*{-48pt}
       \ifx\@articleType\@empty\else{\fontsize{10}{12}\scshape\selectfont\hspace{8pt}\@articleType\hfill\mbox{}\par\vspace{2pt}}\fi%
       \minipage{\linewidth}
         \hrulefill\\[-0.7pt]%
         \parbox{.1\linewidth}{\includegraphics[width=4.2cm]{pharmasprings_logo.png}}\hfill
%         \fcolorbox{kwdboxcolor}{kwdboxcolor}{\parbox{.792\linewidth}{%
         \fcolorbox{kwdboxcolor}{kwdboxcolor}{\parbox{.729\linewidth}{%
          \begin{center}
          \vskip-4pt 
          \scshape\fontsize{18}{20}\selectfont\journalTitle\hfill\end{center}%
          \vspace*{-16pt}{\vspace*{-3.1pt}\fontsize{8}{9}\selectfont Published by Pharma Springs Publication\hfill  Journal Home Page: \href{https://pharmasprings.com/index.php/\jrnlurlname}{\color{blue}\underline{\smash{https://pharmasprings.com/\jrnlurlname}}}}\hspace*{4pt}\mbox{}}}%
       \par\rule{\linewidth}{1.3pt}%
       \endminipage%
    \par\vspace*{9.2pt}\parbox{.98\linewidth}{\linespread{.9}\raggedright\fontsize{14}{17}\selectfont #1}%
    \vspace*{-8pt}%
  }
}


\setlength{\parindent}{0pt}
\setlength{\parskip}{0.4pc plus 1pt minus 1pt}


\fancypagestyle{headings}{%
  \renewcommand{\headrulewidth}{0pt}%
  \renewcommand{\footrulewidth}{0.3pt}
  \fancyhf{}%
  \fancyhead[R]{%
    \fontsize{9.12}{11}\selectfont\RunningAuthor,\ \abbrvJournalTitle\ \@copyrightYear;\space\ifx\@journalVolume\@empty X\else\@journalVolume\fi%
    \ifx\@journalIssue\@empty\else(\@journalIssue)\fi%
    :\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}%
  }%
  \fancyfoot[RE]{\fontsize{9.12}{11}\selectfont\textcopyright\ Pharma Springs Publication\ $|$\space \journalTitle}%
  \fancyfoot[LO]{\fontsize{9.12}{11}\selectfont\textcopyright\ Pharma Springs Publication\ $|$\space \journalTitle}%
  \fancyfoot[RO,LE]{\fontsize{9.12}{11}\selectfont\thepage}
}\pagestyle{headings}


                                                         
\fancypagestyle{plain}{%
  \renewcommand{\headrulewidth}{0pt}%
  \renewcommand{\footrulewidth}{0.3pt}%
  \fancyhf{}%
  \fancyhead[R]{%
    \fontsize{9.12}{11}\selectfont\RunningAuthor,\ \abbrvJournalTitle\ \@copyrightYear;\space\ifx\@journalVolume\@empty X\else\@journalVolume\fi%
    \ifx\@journalIssue\@empty\else(\@journalIssue)\fi%
    :\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}%
  }%
  \fancyfoot[RE]{\fontsize{9.12}{11}\selectfont\textcopyright\ Pharma Springs Publication\ $|$\space \journalTitle}%
  \fancyfoot[LO]{\fontsize{9.12}{11}\selectfont\textcopyright\ Pharma Springs Publication\ $|$\space \journalTitle}%
  \fancyfoot[RO,LE]{\fontsize{9.12}{11}\selectfont\thepage}
  \ifx\@firstPage\@empty\else\setcounter{page}{\@firstPage}\fi
}


\def\NormalBaseline{\def\baselinestretch{1.1}}
\usepackage{textcase}
\setcounter{secnumdepth}{0}

\titleformat{\section}[block]{\bfseries\boldmath\NormalBaseline\filright\fontsize{10.5}{13}\selectfont}
{\thesection}
{6pt}
{\MakeTextUppercase{#1}}
[]
\titleformat{\subsection}[block]{\bfseries\boldmath\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubsection}
{6pt}
{#1}
[]
\titleformat{\subsubsection}[block]{\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubsubsection}
{6pt}
{#1}
[]
\titleformat{\paragraph}[block]{\NormalBaseline\filright\fontsize{10.5}{10}\selectfont}
{\theparagraph}
{6pt}
{#1}
[]
\titleformat{\subparagraph}[block]{\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubparagraph}
{6pt}
{#1}
[]

\titlespacing{\section}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subsection}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subsubsection}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\paragraph}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subparagraph}{0pt}{.5\baselineskip}{.5\baselineskip}

\captionsetup[figure]{skip=1.4pt,font=bf,labelsep=colon,justification=raggedright,singlelinecheck=false}
\captionsetup[table]{skip=1.4pt,font=bf,labelsep=colon,justification=raggedright,singlelinecheck=false}

\def\bibyear#1{#1}
\def\bibjtitle#1{#1}
\def\bibauand{}
\setlength\bibsep{3pt}
\setlength\bibhang{8pt}
\lefthyphenmin = 3
\makeatother
\date{}
\makeatletter
\usepackage{mdframed}
%\usepackage{balance}
\definecolor{unitednationsblue}{RGB}{222,234,246}
\definecolor{cpboxcolor}{RGB}{211,211,211}
\definecolor{citeboxcolor}{RGB}{242,242,242}
\def\printBio{%
  %\noindent\begin{mdframed}[innerbottommargin=7pt,backgroundcolor=unitednationsblue,hidealllines=true,frametitle={ABOUT AUTHORS},innerleftmargin=7pt,leftmargin=2pt,skipabove=1pt,innertopmargin=2pt,innerbottommargin=1pt,skipbelow=1pt]%
  \noindent\begin{mdframed}[innerbottommargin=7pt,backgroundcolor=unitednationsblue,hidealllines=true,frametitle={ABOUT AUTHORS},innerrightmargin=20pt,rightmargin=4,innerleftmargin=7pt,leftmargin=2pt,skipabove=1pt,innertopmargin=2pt,innerbottommargin=1pt,skipbelow=1pt]%
   \@tempcnta=0
    \loop
      \advance \@tempcnta by 1
      \def\aubioCnt{\the\@tempcnta}
      \vskip1pt
%      \noindent\hspace{-2pt}%\fbox{
%     \vspace*{1pt}
      \raisebox{62pt}%{\begin{minipage}[t]{.35\linewidth}%
      \expandafter\ifx\csname authorImage\aubioCnt\endcsname\relax%
       \else%
         \includegraphics[height=60pt,width=60pt]{\csname authorImage\aubioCnt\endcsname}%height=145pt
       \fi%
      %\end{minipage}
      %}%}
      \hskip 4pt%\fbox{
      \raisebox{52pt}{\begin{minipage}[t]{.74\linewidth}%
        {\fontsize{8}{10}\selectfont{\csname authorName\aubioCnt\endcsname\\[5pt]}}
        \fontsize{8}{10}\selectfont\csname authorDetails\aubioCnt\endcsname\\
      \end{minipage}}
%      }
%     \vskip10pt%
    \ifnum\@tempcnta < \theaubio
    \repeat
%     \vspace*{-16.5pt}%
  \end{mdframed}
}


\makeatother


\def\citeauthor{~Jyoshna~M,
              ~Surendra~Babu~S,
              ~Shaik~Aliya,
              ~Syed~Shahbaz~Ali,
              ~Pradeep~Kumar~M}
\def\citetitle{Preparation and Characterization of Natural Face Pack}
\def\copyrightstmt{This is an open access article distributed under the terms of the Creative Commons Attribution-NonCommercial- ShareAlike 4.0 License, which allows others to remix, tweak, and build upon the work non-commercially, as long as the author is credited and the new creations are licensed under the identical terms.}
\publishedDate{}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Feature enabled:
%citation-style: unsrt
%citation-type: numbered
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\usepackage[flushleft]{threeparttablex}

\usepackage{float}

\begin{document}


\def\RunningAuthor{Jyoshna M et al.}
\firstPage{140}
\articleType{Original Article}
\receivedDate{15 Jun 2021}
\acceptedDate{02 Jul 2021}
\revisedDate{01 Jul 2021}
\journalVolume{1}
\journalIssue{3}
\journalDoi{fjphs.v1i3.223}
\copyrightYear{2021}
\eISSN{2583-116X}
\journalDoi{fjphs.v1i3.223}
\def\authorCount{5}
\def\affCount{2}

\def\journalTitle{Future Journal of Pharmaceuticals and Health Sciences}

\title{Preparation and Characterization of Natural Face Pack}
\author{Jyoshna~M\textsuperscript{*}\textsuperscript{1},  
            Surendra~Babu~S\textsuperscript{2},  
            Shaik~Aliya\textsuperscript{2},  
            Syed~Shahbaz~Ali\textsuperscript{2}, 
            Pradeep~Kumar~M\textsuperscript{2}~\\[5pt]{\textsuperscript{1}Department of Pharmaceutical Analysis, Jagan's Institute of Pharmaceutical Sciences, Jangalakandriga (V), Muthukur (M), Nellore-524346, SPSR Nellore (Dist), Andhra Pradesh, India}~\\{\textsuperscript{2}Jagan's Institute of Pharmaceutical Sciences, Jangalakandriga (V), Muthukur (M), Nellore -524346, SPSR Nellore (Dist), Andhra Pradesh, India}}

\begin{abstract}
Herbal Products have huge demand from few decades as they yield good results without any side effects and provide a soothing effect for longer period of time. Using natural ingredients in a cosmetic formulation thus provides healthy and glowing skin. Currently, due to increase in pollution rate, skin allergic conditions, microbes and external effects human skin have become sensitive and prone to faster aging. Here we have made an attempt to formulate a pack ideal for all skin types. Thus, in the current work, the face packs that can be easily made with the available ingredients showed all the benefits and further optimization studies are necessary on various parameters to identify benefits on humans. The flow properties of natural herbal face pack show accurate results. All the ingredients were mixed homogenously and were evaluated for parameters like organoleptic properties (Appearance, color, odor, texture \& smoothness) particle size ranges from 23.5\ensuremath{\pm}2.15 \ensuremath{\mu }m to 24.4\ensuremath{\pm}5.14 \ensuremath{\mu }m, Ash content ranges from 86 \ensuremath{\pm} 0.951 \% to 94 \ensuremath{\pm} 0.34 \%, pH was ranges from 6.51 \ensuremath{\pm} 0.12 to 6.65\ensuremath{\pm} 0.16. Loss on drying ranges from 3.18 \% to 4.12 \%. Irritancy tests, rheological features and stability. The obtained results showed that the prepared formulation was stable on all parameters. Irritancy test revealed negative. Stability tests showed the inertness of the formulation. 
\end{abstract}\def\keywordstitle{Keywords}

\begin{keywords}Herbal Products,\newline Face Pack,\newline Cosmetics,\newline Stability Studies
\end{keywords}
\twocolumn[ \maketitle {\printKwdAbsBox}]
\makeatletter{\raggedright{\Large$^\ast$}Corresponding Author\par Name:\ Jyoshna~M~\\ Phone:\ 9985664874~\\  Email:\ jyoshnagrace@gmail.com
        \par}\vspace*{-13pt}\hrulefill\par\ifx\@eISSN\@empty\else{\fontsize{12}{14}\selectfont eISSN: \@eISSN\hfill pISSN: \@pISSN}\par\fi% 
        \textsc{DOI:}\ \href{https://doi.org/10.26452/\@journalDoi}{\textcolor{blue}{\underline{\smash{https://doi.org/10.26452/\@journalDoi}}}}\par%
        \vspace*{-12pt}\hrulefill\\[5pt]\hypersetup{allcolors=black}\qrcode[version=4,height=1.5cm]{https://doi.org/10.26452/\@journalDoi}\hypersetup{allcolors=blue} \hspace*{10.5pt}\parbox[t]{5cm}{\raisebox{19pt}{\fontsize{9.12}{10.12}\selectfont Production and Hosted by}\\[-13pt]{{\fontsize{12}{14}\selectfont  Pharmasprings.com \\[-1pt]{\fontsize{9.12}{10.12}\selectfont\textcopyright\ \@copyrightYear\ $|$ All rights reserved.}\\[-8pt]\hspace*{-1pt}\rule{.762\columnwidth}{1.2pt}}}}\hypersetup{allcolors=blue}%{\fontsize{12}{14}\selectfont Pharmasprings.com}\par%
        \makeatother
        
    
\section{Introduction}
Herbal cosmetics are also known as natural cosmetics. During the civilization era there were no fairness creams nor the cosmetic surgeries. The objective of this work is preparation and characterization of natural face pack. By varying the concentrations in formulations pertaining the ingredients \unskip~\cite{1141177:22551328}. Herbal cosmetics are the preparation which represent cosmetics that are with active bio-ingredients (Natural), nutraceuticals and pharmaceuticals that are to be applied on our skin. The outermost layer of skin which is exposed to atmosphere is a horny layer i.e. stratum corneum \& is made up of keratinized dead cells. These cells are responsible for maintaining the suppleness of the skin, i.e. maintaining the moisture balance and overall appearance \unskip~\cite{1141177:22551332}.

\textbf{Herbal Cosmetics are Used in an Effective Manner which are Very Much Beneficial like }


  
  \begin{enumerate}
  \item \relax As they are natural, they does not possess any of the harmful effects on the skin 
  \item \relax They are relatively safe than marketed ones
  \item \relax Easily formulated and evaluated depending on their availability
  \item \relax From earlier days results are satisfactory \unskip~\cite{1141177:22551335}.
  \end{enumerate}
  \textbf{Advantages of Applying a Face Pack}

It nourishes the skin. It helps to reduce, pimple, acne, marks and scars based on the ingredients used. Face packs remove dead cells upon the skin. They provide a relaxing and soothing on skin. Face pack also help to restore the lost shine and glow of skin in a short span of time. They help to prevent premature aging of skin \unskip~\cite{1141177:22551333}. By using natural face masks regularly they bring glow to skin and also improves skin texture along with complexion.
    
\section{MATERIALS AND METHODS}
Multani Mitti was purchased from Creative Research Labs Pvt. Ltd. Gujarat. Turmeric Mother Herbs (P) Ltd, New Delhi. Milk Powder to be acquired form Dodla dairy parlour, Nellore. Sandalwood was received as freely given sampling from Red Sandalwood, Naidupeta. Organ peel, Neem, Nutmeg was purchased from Health and beauty shop, Karnataka. All abundant chemical as well as chemical agent utilized in this study in with analytical grade.

\textbf{Methodology}

\textbf{Method of Preparation}

Weigh the finely powdered ingredients accurately with the aid of sieves, preferably sieve \#120 and uniform mixing was done by using serial dilution method \unskip~\cite{1141177:22551330}. By using self-sealable bags preferably polythene, the face pack was packed, labeled and evaluated further [Table~\ref{tw-5ee496d8fd59}].

\textbf{How to apply a face pack? }

Take a bowl and add prepared face pack powder into it as per the required quantity and also add rose water to it and mix until smooth paste is prepared. Apply it to the face and also to the acne region [Figure~\ref{f-4b001eb8fef1}]. Wait for 20-25min untill it becomes dry and finally wash the face with cold water \unskip~\cite{1141177:22551330}.

\textbf{Evaluation of Face Pack\ensuremath{^{}}}

\textbf{Organoleptic Evaluation}

They include odor, nature, color, feel and consistency were evaluated to identify the properties \unskip~\cite{1141177:22551337}. 

\textbf{Physical Evaluation}

The Powder Flow Property of the Natural Herbal face pack as follows

Angle of repose: \ensuremath{\theta } = tan\ensuremath{^{-1}} (h/r) 

Bulk density \ensuremath{\rightarrow} Weight of powder/ Bulk volume

Tapped density \ensuremath{\rightarrow} Weight of powder/ Tapped volume 

Carr's Index (I) = (Tapped Density -  Bulk Density)/(Tapped Density) x100 

Hausner's ratio \ensuremath{\rightarrow} Tapped density/ Bulk density\textbf{\space }

\textbf{Particle Size}

Particle size affects several properties like spreadability, grittiness etc., it is determined by sieving method as per I.P. Standard sieves for 10min by mechanical shaking \unskip~\cite{1141177:22551336}.

\textbf{Ash Content}

Weigh 2-4g of the ground air-dried powder that is previously ignited and tared crucible. Spread the powder evenly and ignite it by increasing the heat to 500-600\ensuremath{^\circ}C gradually until it is white thus indicating the absence of carbon. Cool in a desiccator and weigh \unskip~\cite{1141177:22551334}.

\textbf{pH}

pH of 1\% aqueous solution of the preparation was measured using a calibrated digital pH meter at constant temperature \unskip~\cite{1141177:22551334}.

\textbf{Loss of drying}

Weigh approximately 1.5 gms of the powder into a weighed flat and thin porcelain dish, dry in the oven at 100\ensuremath{^{o}}C or 105\ensuremath{^{o}}C and two consecutive weights must not differ more than 0.5 mg. Cool it in a desiccator and weigh \unskip~\cite{1141177:22551331}. Calculate the loss in weight.

\textbf{Irritancy test}

Irritancy test was conducted in order check whether the prepared formulation showed any irritancy, edema or erythema. This test was repeated at regular intervals by applying on the left-hand dorsal surface for about an area of 1sq.cm approx \unskip~\cite{1141177:22551338}.

\textbf{Stability studies}

Stability testing was done by storing the formulation at temperature conditions for the period of one month \unskip~\cite{1141177:22551329}. 

The packed glass vials were evaluated for physical parameters (Color, Odor, pH, Consistency and feel) at different temperature conditions (35 \ensuremath{\pm} 0.5\ensuremath{^{0}}C to 40 \ensuremath{\pm} 0.5\ensuremath{^{0}}C).

\textbf{RESULTS AND DISCUSSION}

\textbf{Evaluation of Face Pack}

Following evaluations were performed to confirm the ability and superiority of prepared face pack. 


\begin{table*}[!htbp]
\caption{\boldmath {Formulation of Natural Herbal Face Pack} }
\label{tw-5ee496d8fd59}
\centering 
\begin{threeparttable}

\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.07730000000000001\linewidth-2\tabcolsep}p{\dimexpr.1742\linewidth-2\tabcolsep}p{\dimexpr.25740000000000002\linewidth-2\tabcolsep}p{\dimexpr.10919999999999998\linewidth-2\tabcolsep}p{\dimexpr.1471\linewidth-2\tabcolsep}p{\dimexpr.1186\linewidth-2\tabcolsep}p{\dimexpr.11620000000000001\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}\cAlignHack S.No. & \cAlignHack Ingredients & \cAlignHack Scientific name & \multicolumn{4}{p{\dimexpr(.4911\linewidth-2\tabcolsep)}}{\cAlignHack Quantity( 100gms)}\\
\rowcolor{kwdboxcolor}\cAlignHack  & \cAlignHack  & \cAlignHack  & \cAlignHack FP 1 & \cAlignHack FP 2 & \cAlignHack FP 3 & \cAlignHack FP 4\\
\tblmidrule 
\cAlignHack 1 &
  \cAlignHack Multani Mitti &
  \cAlignHack \textit{Calcium bentonite} &
  \cAlignHack 10 &
  \cAlignHack 20 &
  \cAlignHack 25 &
  \cAlignHack 15\\
\cAlignHack 2 &
  \cAlignHack Turmeric &
  \cAlignHack \textit{Curcuma longa} &
  \cAlignHack 20 &
  \cAlignHack 15 &
  \cAlignHack 20 &
  \cAlignHack 25\\
\cAlignHack 3 &
  \cAlignHack Milk Powder &
  \cAlignHack - &
  \cAlignHack 15 &
  \cAlignHack 10 &
  \cAlignHack 15 &
  \cAlignHack 15\\
\cAlignHack 4 &
  \cAlignHack Sandal Wood &
  \cAlignHack \textit{Santalum alba} &
  \cAlignHack 20 &
  \cAlignHack 15 &
  \cAlignHack 15 &
  \cAlignHack 20\\
\cAlignHack 5 &
  \cAlignHack Orange peel &
  \cAlignHack \textit{Citrus reticulate} &
  \cAlignHack 25 &
  \cAlignHack 17 &
  \cAlignHack 6 &
  \cAlignHack 10\\
\cAlignHack 6 &
  \cAlignHack Neem &
  \cAlignHack \textit{Azadirachta indica} &
  \cAlignHack 3 &
  \cAlignHack 18 &
  \cAlignHack 15 &
  \cAlignHack 10\\
\cAlignHack 7 &
  \cAlignHack Nutmeg &
  \cAlignHack \textit{Myristica fragrans } &
  \cAlignHack 7 &
  \cAlignHack 5 &
  \cAlignHack 4 &
  \cAlignHack 5\\
\tblbottomrule 
\end{tabulary}\par 
\begin{tablenotes}\footnotesize 
    
\item{*FP: Face Pack}
\end{tablenotes}
\end{threeparttable}

\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Organoleptic Properties natural herbal face pack} }
\label{tw-b2d55bbaac12}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.09080000000000002\linewidth-2\tabcolsep}p{\dimexpr.17250000000000004\linewidth-2\tabcolsep}p{\dimexpr.1766\linewidth-2\tabcolsep}p{\dimexpr.20009999999999998\linewidth-2\tabcolsep}p{\dimexpr.16\linewidth-2\tabcolsep}p{\dimexpr.20\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}\cAlignHack S. No. & \cAlignHack Parameters & \multicolumn{4}{p{\dimexpr(.7367\linewidth-2\tabcolsep)}}{\cAlignHack Observation}\\
\rowcolor{kwdboxcolor}\cAlignHack  & \cAlignHack  & \cAlignHack FP 1 & \cAlignHack FP 2 & \cAlignHack FP 3 & \cAlignHack FP 4\\
\tblmidrule 
\cAlignHack 1 &
  \cAlignHack Appearance &
  \cAlignHack Free Flowing powder &
  \cAlignHack Free Flowing powder &
  \cAlignHack Free Flowing powder &
  \cAlignHack Free Flowing powder\\
\cAlignHack 2 &
  \cAlignHack Color &
  \cAlignHack Slight Yellow &
  \cAlignHack Slight yellow &
  \cAlignHack yellow &
  \cAlignHack Slight Yellow\\
\cAlignHack 3 &
  \cAlignHack Odor &
  \cAlignHack Mild &
  \cAlignHack Mild &
  \cAlignHack Mild &
  \cAlignHack Mild\\
\cAlignHack 4 &
  \cAlignHack Texture &
  \cAlignHack Fine &
  \cAlignHack Fine &
  \cAlignHack Fine &
  \cAlignHack Fine\\
\cAlignHack 5 &
  \cAlignHack Smoothness &
  \cAlignHack Smooth &
  \cAlignHack Smooth &
  \cAlignHack Smooth &
  \cAlignHack Smooth\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Evaluation of flow properties of natural herbal face pack} }
\label{tw-adf54f957829}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.09630000000000003\linewidth-2\tabcolsep}p{\dimexpr.3316\linewidth-2\tabcolsep}p{\dimexpr.1323\linewidth-2\tabcolsep}p{\dimexpr.1282\linewidth-2\tabcolsep}p{\dimexpr.1559\linewidth-2\tabcolsep}p{\dimexpr.1557\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}\cAlignHack S.No. & \cAlignHack Parameter & \cAlignHack FP 1 & \cAlignHack FP 2 & \cAlignHack FP 3 & \cAlignHack FP 4\\
\tblmidrule 
\cAlignHack 1 &
  \cAlignHack Tapped density (g/cm\ensuremath{^{3}}) &
  \cAlignHack 1.298 &
  \cAlignHack 1.173 &
  \cAlignHack 1.111 &
  \cAlignHack 1.152\\
\cAlignHack 2 &
  \cAlignHack Bulk density (g/cm\ensuremath{^{3}}) &
  \cAlignHack 1.00 &
  \cAlignHack 1.00 &
  \cAlignHack 1.09 &
  \cAlignHack 1.03\\
\cAlignHack 3 &
  \cAlignHack Hausner's ratio (H) &
  \cAlignHack 1.231 &
  \cAlignHack 1.123 &
  \cAlignHack 1.023 &
  \cAlignHack 1.083\\
\cAlignHack 4 &
  \cAlignHack Angle of repose (\ensuremath{\theta }) &
  \cAlignHack 31.520 &
  \cAlignHack 30.140 &
  \cAlignHack 31.356 &
  \cAlignHack 31.908\\
\cAlignHack 5 &
  \cAlignHack Carr's index (I) &
  \cAlignHack 20.82\% &
  \cAlignHack 21.21\% &
  \cAlignHack 21.59\% &
  \cAlignHack 22.32\%\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Physicochemical Evaluation of natural herbal face pack} }
\label{tw-c93002c1e0e1}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.0909\linewidth-2\tabcolsep}p{\dimexpr.2803\linewidth-2\tabcolsep}p{\dimexpr.142\linewidth-2\tabcolsep}p{\dimexpr.15579999999999998\linewidth-2\tabcolsep}p{\dimexpr.15590000000000002\linewidth-2\tabcolsep}p{\dimexpr.17509999999999998\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}\cAlignHack Sr. No. & \cAlignHack Parameters & \multicolumn{4}{p{\dimexpr(.62879999999999995\linewidth-2\tabcolsep)}}{\cAlignHack Observation}\\
\rowcolor{kwdboxcolor}\cAlignHack  & \cAlignHack  & \cAlignHack FP 1 & \cAlignHack FP 2 & \cAlignHack FP 3 & \cAlignHack FP 4\\
\tblmidrule 
\cAlignHack 1 &
  \cAlignHack Particle size (\ensuremath{\mu }m) &
  \cAlignHack 24.4\ensuremath{\pm}5.14 &
  \cAlignHack 23.5\ensuremath{\pm}2.15 &
  \cAlignHack 24.2\ensuremath{\pm}3.16 &
  \cAlignHack 24.4\ensuremath{\pm}5.11\\
\cAlignHack 2 &
  \cAlignHack Ash content &
  \cAlignHack 91 \ensuremath{\pm} 0.182 &
  \cAlignHack 86 \ensuremath{\pm} 0.951 &
  \cAlignHack 94 \ensuremath{\pm} 0.345 &
  \cAlignHack 91 \ensuremath{\pm} 0.319\\
\cAlignHack 3 &
  \cAlignHack pH &
  \cAlignHack 6.51 \ensuremath{\pm} 0.12 &
  \cAlignHack 6.61 \ensuremath{\pm} 0.26 &
  \cAlignHack 6.65\ensuremath{\pm} 0.16 &
  \cAlignHack 6.61\ensuremath{\pm} 0.11\\
\cAlignHack 4 &
  \cAlignHack Loss on Drying &
  \cAlignHack 4.12 &
  \cAlignHack 3.18 &
  \cAlignHack 4.11 &
  \cAlignHack 3.13\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Irritancy Test of natural herbal face pack} }
\label{tw-038c87d2660c}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.0771\linewidth-2\tabcolsep}p{\dimexpr.17439999999999998\linewidth-2\tabcolsep}p{\dimexpr.14119999999999997\linewidth-2\tabcolsep}p{\dimexpr.1377\linewidth-2\tabcolsep}p{\dimexpr.14\linewidth-2\tabcolsep}p{\dimexpr.14120000000000001\linewidth-2\tabcolsep}p{\dimexpr.1884\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}\cAlignHack S. No. & \cAlignHack Evaluation & \multicolumn{4}{p{\dimexpr(.56010000000000005\linewidth-2\tabcolsep)}}{\cAlignHack Formulations} & \cAlignHack Observation \\
\rowcolor{kwdboxcolor}\cAlignHack  & \cAlignHack  & \cAlignHack FP 1  & \cAlignHack FP 2 & \cAlignHack FP 3 & \cAlignHack FP 4 & \cAlignHack \\
\tblmidrule 
\cAlignHack 1 &
  \cAlignHack Irritant &
  \cAlignHack Present &
  \cAlignHack Absent &
  \cAlignHack Present &
  \cAlignHack Present &
  \cAlignHack Irritation Absent\\
\cAlignHack 2 &
  \cAlignHack Erythema &
  \cAlignHack Absent  &
  \cAlignHack Absent  &
  \cAlignHack Absent  &
  \cAlignHack Absent &
  \cAlignHack Irritation Absent\\
\cAlignHack 3 &
  \cAlignHack Edema &
  \cAlignHack Absent  &
  \cAlignHack Absent  &
  \cAlignHack Absent  &
  \cAlignHack Absent &
  \cAlignHack Irritation Absent\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Parameters of Stability studies of Formulation F2 of natural herbal face pack} }
\label{tw-bd5cd5e54af1}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.1221\linewidth-2\tabcolsep}p{\dimexpr.19\linewidth-2\tabcolsep}p{\dimexpr.26470000000000002\linewidth-2\tabcolsep}p{\dimexpr.18009999999999998\linewidth-2\tabcolsep}p{\dimexpr.2431\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}\cAlignHack S. No. & \cAlignHack Parameters & \multicolumn{3}{p{\dimexpr(.6879\linewidth-2\tabcolsep)}}{\cAlignHack Observations (Formulations FP 2)}\\
\rowcolor{kwdboxcolor}\cAlignHack  & \cAlignHack  & \cAlignHack Room Temperature & \cAlignHack 35\ensuremath{\pm}0.5\ensuremath{^{0}}C & \cAlignHack 40\ensuremath{\pm}0.5\ensuremath{^{0}}C\\
\tblmidrule 
\cAlignHack 1 &
  \cAlignHack Color &
  \cAlignHack No variation &
  \cAlignHack No variation &
  \cAlignHack No variation\\
\cAlignHack 2 &
  \cAlignHack Odour &
  \cAlignHack No variation &
  \cAlignHack No variation &
  \cAlignHack Slightly variation\\
\cAlignHack 3 &
  \cAlignHack pH &
  \cAlignHack 6.61 \ensuremath{\pm} 0.26 &
  \cAlignHack 6.61 \ensuremath{\pm} 0.26 &
  \cAlignHack 7.01 \ensuremath{\pm}0.29\\
\cAlignHack 4 &
  \cAlignHack Texture &
  \cAlignHack Soft and fine &
  \cAlignHack Soft and fine &
  \cAlignHack Soft and fine\\
\cAlignHack 5 &
  \cAlignHack Smoothness &
  \cAlignHack Very Smooth &
  \cAlignHack Very Smooth &
  \cAlignHack Very Smooth\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\bgroup
\fixFloatSize{images/b15d787d-1e12-4f84-a45a-f390a8a0f74d-upicture1.png}
\begin{figure*}[!htbp]
\centering \makeatletter\IfFileExists{images/b15d787d-1e12-4f84-a45a-f390a8a0f74d-upicture1.png}{\includegraphics[width=.86\linewidth]{images/b15d787d-1e12-4f84-a45a-f390a8a0f74d-upicture1.png}}{}
\makeatother 
\caption{\boldmath {Packing of Herbal Face Pack}}
\label{f-4b001eb8fef1}
\end{figure*}
\egroup
\textbf{Organoleptic Evaluation }

Morphological parameters were evaluated and depicted in Table~\ref{tw-b2d55bbaac12}. Colour of formulation was Slight yellow. 

The odour of formulations was pleasant and acceptable. Texture and smoothness was also acceptable as per requirement of natural Herbal formulations.

\textbf{Powder Flow Property}

The prepared face pack was evaluated for flow properties and are given in the Table~\ref{tw-adf54f957829}.

\textbf{Physico chemical Evaluation}

Herbal face pack was prepared and evaluated for physicochemical parameters were showed in the Table~\ref{tw-c93002c1e0e1}. The pH of formulation was found close to neutral. The ash content and moisture content were found to be within limit. The particle size of prepared formulations was found in the range of 23.5 \ensuremath{\pm} 2.5 \ensuremath{\mu }m.

\textbf{Irritancy Test}

The evaluation report of irritancy test was shown in the Table~\ref{tw-038c87d2660c}. The formulation showed absence of irritation, redness and swelling. The prepared formulation was safe to use on skin.

\textbf{Stability Studies }

The results of stability were shown in Table~\ref{tw-bd5cd5e54af1}. No change in color, odour, texture and smoothness was observed at mentioned conditions of stability except pH.
    
\section{CONCLUSION}
In the current scenario, people are in search of cure for numerous skin problems without harmful side effects. Natural/ Herbal ingredients proved itself in formulating the cosmetics without any side effects to the skin. These face packs are preferred for sustaining and effective way to increase the glow and appearance of skin. In the current work, we have made an attempt to formulate the herbal face pack that contains natural ingredients namely multani mitti, turmeric, milk powder, sandalwood, neem orange peel and nutmeg. The prepared herbal formulation was physico-chemically and microbiologically stable, and also possessed characteristics of a standard formulation for applying on skin.
    
\section{Acknowledgement}
The authors are thankful to the Secretary \& Correspondent (Mr. Madhu Sudhan Reddy Sir) \& Principal (Dr. P. Venkatesh Sir) of Jagan's Institute of Pharmaceutical Sciences, Jangalakandriga (V), Muthukur (M) Nellore-524 346, SPSR Nellore (Dist) Andhra Pradesh for providing the necessary facilities to carry out this work.

\textbf{Conflict of interest}

The authors attest that they have no conflict of interest in this study.

\textbf{Funding support}

No financial support for the current study.
    

\bibliographystyle{unsrt}

\bibliography{\jobname}\vspace*{20pt} 
\begin{mdframed}[backgroundcolor=cpboxcolor,hidealllines=true,innerbottommargin=7pt,innerrightmargin=1pt,rightmargin=1pt,innerleftmargin=1pt,leftmargin=2pt,skipabove=1pt,innertopmargin=2pt,innerbottommargin=1pt,skipbelow=1pt]%
{\fontsize{8.5}{10}\selectfont\textbf{Copyright:}~\copyrightstmt}	
\end{mdframed}
        
\begin{mdframed}[backgroundcolor=citeboxcolor,hidealllines=true,innerbottommargin=7pt,innerrightmargin=4pt,rightmargin=4pt,innerleftmargin=2pt,leftmargin=2pt,skipabove=1pt,innertopmargin=2pt,innerbottommargin=4pt,skipbelow=2pt]%
{\fontsize{8.5}{10}\selectfont\textbf{Cite~this~article:}~\citeauthor.~\citetitle.\space\makeatletter\abbrvJournalTitle\ \@copyrightYear;\ \ifx\@journalVolume\@empty X\else\@journalVolume\fi\ifx\@journalIssue\@empty\else(\@journalIssue)\fi:\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}.\makeatother}	
\end{mdframed}
\vskip-1.1pc

\vskip.5pc
{\includegraphics[width=5cm]{pharmasprings_logo.png}}
~\\
{\fontsize{9.12}{10.12}\selectfont\textcopyright\ \makeatletter\@copyrightYear\makeatother\ Pharma Springs Publication.}

\end{document}
